Author's response to reviews

Title: Subgroup effects despite homogeneous heterogeneity test results

Authors:

   Rolf HH Groenwold (r.h.h.groenwold@umcutrecht.nl)
   Maroeska M Rovers (M.Rovers@umcutrecht.nl)
   Jacobus Lubsen (j.lubsen@compuserve.com)
   Geert JMG van der Heijden (G.vanderHeijden@umcutrecht.nl)

Version: 4 Date: 11 May 2010

Author's response to reviews: see over
Dear BMC Series Editorial Production Team,

Please find attached the revised version of our manuscript “Subgroup effects despite homogenous heterogeneity test results” (MS 1198767956350389) for publication in BMC Medical Research Methodology.

We have changed the structure of the manuscript as requested. Furthermore, in the revised version of the manuscript we cited each figure and we included legends for each figure as well. Also, a statement is included to the authors’ contributions section that ‘all authors read and approved the final manuscript’. Finally, we included the email addresses of all authors on the title page.

Yours sincerely,

On behalf of all authors,

Rolf H.H. Groenwold, MD PhD